Author's response to reviews

Title: Lymph node removal enhances corneal graft survival in mice at high risk of rejection

Authors:

Dr Jarmila Plskova (plskovaj@hotmail.com)
Vladimir Holan (holan@img.cas.cz)
Dr Martin Filipec (martin.filipec@vfn.cz)
Prof John V Forrester (j.forrester@abdn.ac.uk)

Version: 5 Date: 22 Mar 2004

The changes, which has been made to the manuscript:
1. The Abstract starts now on the second page of the manuscript and not on the title page.
2. Competing interests were formated as requested by deleting repetition of 'competing interests'.
3. Author's contributions were changed: each author's name was replaced by initials.